

Author's response to reviews

Title: Targeted rotavirus vaccination of high-risk infants; a low cost and highly cost-effective alternative to universal vaccination

Authors:

Patricia Bruijning-Verhagen (p.bruijning@umcutrecht.nl)
Marie-Josée J Mangen (M.J.J.Mangen@umcutrecht.nl)
Mariet Felderhof (mfelderhof@spaarneziekenhuis.nl)
Nico G Hartwig (n.hartwig@erasmusmc.nl)
Marlies van Houten (MvanHouten@spaarneziekenhuis.nl)
Léon Winkel (L.winkel@kg.nl)
Wouter J de Waal (wdwaal@diakhuis.nl)
Marc JM Bonten (M.J.M.Bonten@umcutrecht.nl)

Version: 4 **Date:** 28 March 2013

Author's response to reviews:

To the editors,

Thank you for accepting the revisions made to our manuscript "Targeted rotavirus vaccination of high-risk infants; a low cost and highly cost-effective alternative to universal vaccination". The additional editorial changes have been made as requested in the newly uploaded manuscript file. See below for further details.

Sincerely,

Patricia Bruijning-Verhagen
corresponding author

Editorial changes requested:

1. Competing interests: Please add the heading "Competing interests" to the section starting with "All authors declare no financial support ..."

Added to text

2. Please add a list of 3–10 keywords after the abstract to describe the main contents of the manuscript

Keywords added

3. Referee 2's comment at round 1 of review: "Similarly, you should indicate and name the hospitals involved in RV active surveillance in 2011..."

Your response: Added to the legend of Table S4

In the version of the manuscript we have, we are not able to see any information

on the name of the hospitals in the legend of table S4. Please could you submit a version of the legend with this information included?

Hospital names were added to the legend of table S3 and not table S4. Sorry for the confusion.